Author’s response to reviews

Title: Anti-hyperalgesic properties of a flavanone derivative Poncirin in acute and chronic inflammatory pain models in mice

Authors:

Ruqayya Afridi (r.afridi36@gmail.com)
Ashrafullah Khan (ashrafwazir6@gmail.com)
Sidra Khalid (sidra.merlin@gmail.com)
Busha Shal (bushra.shal@gmail.com)
Hina Rasheed (hinarasheed137@gmail.com)
Zia Ullah (ziaullah121@gmail.com)
Omer Shehzad (omersnu@yahoo.com)
Yeong Shik Kim (kims@snu.ac.kr)
Salman Khan (skhan@qau.edu.pk)

Version: 1 Date: 30 Jun 2018

Author’s response to reviews:

1. Please provide more justification for the contributions of authors. Currently, the contributions of HA and OS do not automatically qualify for authorship. Please amend your manuscript accordingly.

Response: All the authors contribution were fully addressed in the revised manuscript.

2. If two authors have the same initials, please distinguish between them by designating them with numbers 1 and 2. For example: SK1 and SK2. SK1 would correspond to the author furthest up on the author list.

Response: As per recommendation of editor, both the authors (SK1 and Sk2) was differentiated by mentioning Skhalid and Sk.
3. Please add a "Consent for publication" section in the Declarations. You may state "Not applicable" in this section as no personal information is provided in your manuscript.

Response: "Consent for publication" was added in the revised manuscript in Declaration part.

4. Please change the heading for this section from “Ethics Statement” to “Ethics approval and consent to participate”.

Response: “Ethics Statement” was changed to “Ethics approval and consent to participate”.

5. Please ensure that all figures are uploaded as individual files in the file inventory. When uploading the figure files, please make sure they contain the image graphic only and not the titles or captions.

Response: All the figures were uploaded separately as per recommendation of editor.